Daily Status 25 October

10/25/13
Summary:

® No IOP last night
® mostly clear this morning, few clouds to the North
® 40m TRH on rim went out last night

Actions past 24 hours:
® SO added a Qsoil probe to base for testing
To dos:

® SO to water new Qsoil probe at base
® switch pocketec

Sensor status:

T/RH: ok, monitor 40m TRH on rim
P: ok

csat u,v: ok

csat Idiag: ok

csat w, tc: ok

kh2o: ok

motes: ok

Wetness: ok

radiation: ok

Tsoil: ok

Gsoil: ok

Qsaoil: ok, monitor Qsoil at base since ground has been watered
Cvsoil: ok.

2D sonic: ok
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